Author's response to reviews

Title: miR-372 down-regulates the oncogene ATAD2 to influence hepatocellular carcinoma proliferation and metastasis

Authors:

Gang Wu (wugzwl@gmail.com)
Haiyang Liu (liuhaiy23@sohu.com)
Hui He (hethm@gmail.com)
Yawei Wang (yawei87@sina.com)
Xiaojun Lu (xiaojunl@sohu.com)
Yanqiu Yu (yanqiu68@gmail.com)
Shuguan Xia (shuguanxia@sohu.com)
Xiangyu Meng (53242347@qq.com)
Yongfeng Liu (Yongfeng52@126.com)

Version: 4 Date: 5 January 2014

Author's response to reviews: see over
Response to comments of editor
Ms. No.: 1874830003109180.
Title: miR-372 down-regulates the oncogene ATAD2 to influence hepatocellular carcinoma proliferation and metastasis

Dear editor,

We have added abstract in the manuscript. And at the end of the manuscript, the author's contribution and competing interest section were added after Acknowledgements.

Thank you for considering publishing our manuscript. We will work harder in further study.

Best wishes

Gang Wu